Author’s response to reviews

Title: The Resonance® metallic ureteral stent in the treatment of malignant ureteral obstruction: A prospective observational study

Authors:

Jun Miyazaki (jmiyazaki@iuwh.ac.jp)
Mizuki Onozawa (mizukionozawa@iuwh.ac.jp)
Satoshi Takahashi (stakahas@sapmed.ac.jp)
Yuka Maekawa (yuka_mae@gifu-u.ac.jp)
Mitsuru Yasuda (super7@gifu-u.ac.jp)
Koichiro Wada (gmd17055@s.okadai.jp)
Yuji Maeda (y-maeda@mattohp.com)
Takuro Masaki (t-masaki@harasanshin.or.jp)
Akito Yamaguchi (yamaguchi@harasanshin.or.jp)
Masahiko Suzuki (Masahiko12o11u08@aol.com)
Yasuyuki Sakai (yasakai@cancer-c.pref.saitama.jp)
Tomokazu Kimura (tomokazu.kim@md.tsukuba.ac.jp)
Manabu Takai (manabu11gaku@yahoo.co.jp)
Kensaku Seike (k.seike-04.6.26@leaf.ocn.ne.jp)
Takahiko Hashimoto (th-uro@hyo-med.ac.jp)
Shingo Yamamoto (shingoy@hyo-med.ac.jp)

Version: 3 Date: 06 Dec 2019

Author’s response to reviews:

1. Please could you format your Author’s contributions in a list, rather than a table. Please keep the contributions as individual initials and include the statement the state that all authors read and approved the final manuscript. An example of an acceptable format can be found here: https://bmcurology.biomedcentral.com/articles/10.1186/s12894-019-0543-8

We formatted our Author’s contribution in a list.
2. Please provide more details regarding the specific role of each of the funding bodies in your study. If the funding bodies only provided the financial means to allow the authors to carry out the study, please state “The funding bodies played no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript” and include this statement in the Funding section of your manuscript.

However, if each of the funding bodies took part in the design of the study and collection, analysis, and interpretation of data, and the writing of the manuscript, this must be stated in the Competing Interests section of your manuscript.

We provided funding bodies in our study. We added “The funders had no role in study design, study completion, data analysis, and writing of the manuscript. “

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Please ensure all figures, tables and additional files are cited in the text, as they may be removed by the production department. Please delete any additional files not intended for publication.

We uploaded a single, final, clean version.

Reviewer reports:
Noor Peter N. Buchholz (Reviewer 1): Innovative article with good datasets
Thank you for your comment. We appreciate.